Author's response to reviews

Title: Measuring access to medicines: a review of quantitative methods used in household surveys

Authors:

Vera Maria V Paniz (vpvieira@terra.com.br)
Anaclaudia G Fassa (anaclaudia.fassa@gmail.com)
Maria de Fátima S Maia (fatima-maia@uol.com.br)
Marlos R Domingues (coriolis@vetorial.net)
Andréa D Bertoldi (andreabetoldi@terra.com.br)

Version: 7 Date: 21 May 2010

Author's response to reviews: see over
Author's response to BMC Series Editorial Production Team

Title: Measuring access to medicines: a review of quantitative methods used in household surveys

Journal: BMC Health Services Research

MS ID: 3334901592936810

Authors:
Vera Maria V Paniz
Anaclaudia G Fassa
Maria de Fátima S Maia
Marlos R Domingues
Andréa D Bertoldi

Date: 21 May 2010

Author's response:
We corrected the Acknowledgments in the body of the text.